Author's response to reviews

Title: No changes of cholesterol levels with a commercially available glucosamine product in patients treated with lipid lowering drugs a controlled, randomised, open cross-over trial

Authors:

Robert Eggertsen (robert.eggertsen@vgregion.se)
Åke Andreasson (ake.andreasson@vgregion.se)
Lennart Andrén (lennart.andren@vgregion.se)

Version: 5 Date: 3 August 2012

Author's response to reviews:

Cover letter to manuscript MS: 2537355160657648, No changes of cholesterol levels…

Thank you for further comments to which we have made the following appropriate changes in the manuscript:

Ref Dr Andrew Tershakovec

1: We agree and have corrected this regarding the atorvastatin group in the title, abstract conclusion, results, discussion and the final conclusion.

2: It maybe a short period for HbA1c but in clinical use the analysis is used for periods of 1-3 months. We think it is of interest to show the outcome. However, we have mentioned this in the discussion.

3. We think as Dr Tershakovec suggests that similar result will be obtained with other glucosamine products. We have corrected this in title, abstract, end of introduction and in discussion.

4. This is corrected in Table 1.

Regarding English language some emendations have been made.